Author's response to reviews

Title: Chronic cerebrospinal venous insufficiency in Multiple Sclerosis: an highly prevalent age-dependent phenomenon without clinical correlations

Authors:

Roberta Lanzillo (roberta.lanzillo@unina.it)
Marcello Mancini (marcello.mancini58@gmail.com)
Raffaele Liuzzi (raffaele.liuzzi@cnr.it)
Orlando Di Donato (orlandodd@hotmail.com)
Elena Salvatore (elena.salvatore@unina.it)
Valentina Maglio (valentina.maglio@gmail.com)
Giovanni Vacca (centrosm@unina.it)
luca Amato (centrosm@unina.it)
Gennaro D'Anna (gennaro.danna@gmail.com)
Arturo Brunetti (arturo.brunetti@unina.it)
Vincenzo Brescia Morra (bresciamorra@unina.it)

Version: 2 Date: 22 August 2012

Author's response to reviews:

Dear Nikka Portodo
Journal Editorial Office
BioMed Central

Subject: MS: 3697861107767469- Pre-review requests

MS: 3697861107767469
Research article
Chronic cerebrospinal venous insufficiency in Multiple Sclerosis: an highly prevalent age-dependent phenomenon without clinical correlations
Roberta Lanzillo, Marcello Mancini, Raffaele Liuzzi, Orlando Di Donato, Elena Salvatore, Valentina Maglio, Giovanni Vacca, luca Amato, Gennaro D'Anna, Arturo Brunetti and Vincenzo Brescia Morra
BMC Neurology (Section: Demyelinating diseases)

Dear Nikka,

I made all the required changes, I hope that now the manuscript is correct
Below a point-by-point description of the changes made.

1) Please provide separate emails for all authors in both the submission system and the manuscript.
I have provided separate emails for all authors in both submission system and the manuscript

Roberta Lanzillo (roberta.lanzillo@unina.it)
Marcello Mancini (marcello.mancini58@gmail.com)
Raffaele Liuzzi (raffaele.liuzzi@cnr.it)
Orlando Di Donato (orlandodd@hotmail.com)
Elena Salvatore (elena.salvatore@unina.it)
Valentina Maglio (valentina.maglio@gmail.com)
Giovanni Vacca (giovagv@libero.it)
Luca Amato (amatocir@tiscali.it)
Gennaro D'Anna (gennaro.danna@gmail.com)
Arturo Brunetti (arturo.brunetti@unina.it)
Vincenzo Brescia Morra (bresciamorra@unina.it)

(2) Acknowledgments:

Anyone other than the authors of the article contributed to the work and should be included in the acknowledgments section.

(3) Authors Contributions:

In order to give appropriate credit to each author of a paper, I have specified the individual contributions of authors and this has been added to the manuscript:

L.R. was involved in conception and design, acquisition of data, analysis and interpretation of data and manuscript drafting
M.M. was involved in study concepts and design, data acquisition, data analysis and interpretation, manuscript drafting, literature research
L.R. performed statistical analysis and participated to interpretation of data and manuscript drafting
D.D.O. participated to conceived of the study, and participated in its design and coordination and helped to draft the manuscript
S.E. conceived of the study, and participated in its design and coordination
M.V.: participated in data acquisition, manuscript editing data acquisition, literature research
V.G. literature research and data interpretation
A.L. literature research and data interpretation
D.G.: data acquisition, data analysis/interpretation
B.A.: was involved in conception and design of the study
B.M.V. was involved in conception and design of the study, in data analysis and
interpretation
All authors read and approved the final manuscript.

Sincerely
Raffaele Liuzzi, PhD
Institute of Biostructure and Bioimaging.
National Research Council Naples, Italy